\begin{table}%t3
\caption{\label{tab3}Other lines studied.}
%\centering
\small
\begin{tabular}{ccccccc}
\hline\hline
%&&&&&& \\[-8pt]
    & & & & &\multicolumn{2}{c}{Equivalent widths (m\AA)} \\
\cline{6 -7} 
%&&&&&& \\[-8pt]
               &   &   &  &                       &   HD   &  BD+04$^\circ$  \\ 
$\lambda$~(\AA) & Species  & $\chi$~(eV) & $gf$ & Ref & 10613  & 2466 \\
\hline
5682.65 & Na~{\sc i} & 2.10 & 1.995e-01 &  PS & 50 & 10 \\
5688.22 & Na~{\sc i} & 2.10 & 3.981e-01 &  PS & 77 & 20 \\
8717.83 & Mg~{\sc i} & 5.91 & 1.956e-01 & WSM & 37 & ---\\
8736.04 & Mg~{\sc i} & 5.94 & 4.571e-01 & WSM & 67 & ---\\
5793.08 & Si~{\sc i} & 4.93 & 8.600e-03 & R03 & 30 & 10 \\
6125.03 & Si~{\sc i} & 5.61 & 2.884e-02 & E93 & 21 & ---\\
6145.02 & Si~{\sc i} & 5.61 & 3.715e-02 & E93 & 18 & ---\\
6155.14 & Si~{\sc i} & 5.62 & 1.698e-01 & E93 & 47 & 13 \\
7800.00 & Si~{\sc i} & 6.18 & 1.905e-01 & E93 & 24 & ---\\
8742.45 & Si~{\sc i} & 5.87 & 3.090e-01 & E93 & 63 & ---\\
6102.73 & Ca~{\sc i} & 1.88 & 1.621e-01 & D2002 & 123 & 63 \\
6122.23 & Ca~{\sc i} & 1.89 & 4.786e-01 & D2002 & --- & 95 \\
6161.30 & Ca~{\sc i} & 2.52 & 5.370e-02 & E93   & 71  & 16 \\
6162.18 & Ca~{\sc i} & 1.90 & 8.128e-01 & D2002 & --- & 107 \\
6166.44 & Ca~{\sc i} & 2.52 & 7.244e-02 & R03  & 68 & 13 \\
6169.04 & Ca~{\sc i} & 2.52 & 1.585e-01 & R03  & 83 & 39 \\
6169.56 & Ca~{\sc i} & 2.53 & 3.311e-01 & DS91 & 98 & 39 \\
6439.08 & Ca~{\sc i} & 2.52 & 2.951e+00 & D2002 & 147 & 92 \\
6455.60 & Ca~{\sc i} & 2.51 & 5.129e-02 & R03 & 57 & 33 \\
6471.66 & Ca~{\sc i} & 2.51 & 2.041e-01 & ~S86 & 91 & 33 \\
6493.79 & Ca~{\sc i} & 2.52 & 7.762e-01 & DS91 & 120 & 60 \\
5239.82 & Sc~{\sc ii} & 1.45 & 1.698e-01 & MFW & 77 & 39 \\
5526.82 & Sc~{\sc ii} & 1.77 & 1.318e+00 & MFW & 93 & 58 \\
5657.88 & Sc~{\sc ii} & 1.51 & 2.884e-01 & GS  & 78 & 43 \\
6245.62 & Sc~{\sc ii} & 1.51 & 9.555e-01 & R03 & --- & 15 \\
6604.60 & Sc~{\sc ii} & 1.36 & 5.012e-02 & R03 & 53  & 19 \\
4533.25 & Ti~{\sc i} &  0.85 & 3.388e+00 & D2002 & --- & 69 \\
4534.78 & Ti~{\sc i} & 0.84 & 1.905e+00 & D2002 & 111 & 57 \\
4981.72 & Ti~{\sc i} & 0.84 & 3.162e+00 & D2002 & 139 & --- \\
5087.06 & Ti~{\sc i} & 1.43 & 1.445e-01 & E93 & 44 & ---\\
5113.45 & Ti~{\sc i} & 1.44 & 1.318e-01 & E93 & 35 & ---\\
5866.46 & Ti~{\sc i} & 1.07 & 1.345e-01 & E93 & 67 & ---\\
6091.18 & Ti~{\sc i} & 2.27 & 4.266e-01 & R03 & 64 & ---\\
6126.22 & Ti~{\sc i} & 1.05 & 4.266e-02 & R03 & 135 & ---\\
6261.10 & Ti~{\sc i} & 1.43 & 3.311e-01 & B86 & 66 & --- \\
5296.70 & Cr~{\sc i} & 0.98 & 4.074e-02 & GS & 103 & 28 \\
5300.75 & Cr~{\sc i} & 0.98 & 7.413e-03 & GS & 66 & --- \\
5345.81 & Cr~{\sc i} & 1.00 & 1.047e-01 & GS & 140& 50 \\
5348.33 & Cr~{\sc i} & 1.00 & 5.129e-02 & GS & 109& 28 \\
5409.80 & Cr~{\sc i} & 1.03 & 1.905e-01 & GS & ---&  60 \\
6330.09 & Cr~{\sc i} & 0.94 & 1.202e093 & R03 & 37 & --- \\
5084.11 & Ni~{\sc i} & 3.680 & 6.607e-01 & E93 & 72 & ---\\
5115.40 & Ni~{\sc i} & 3.834 & 5.248e-01 & R03 & 55 & ---\\
6176.82 & Ni~{\sc i} & 4.088 & 5.450e-01 & R03 & 44 & ---\\
6327.60 & Ni~{\sc i} & 1.68 & 7.709e-04 & MFW & 46  & ---\\
6482.80 & Ni~{\sc i} & 1.94 & 2.344e-02 & MFW & 44  & ---\\
6586.33 & Ni~{\sc i} & 1.95 & 1.549e-03 & MFW & 39  & ---\\
6643.64 & Ni~{\sc i} & 1.68 & 9.332e-03 & MFW & 100 & 24 \\
6767.77 & Ni~{\sc i} & 1.83 & 6.761e-03 & MFW & 88 & 23 \\
6772.32 & Ni~{\sc i} & 3.66 & 1.072e-01 & R03 & 35 & 15 \\
7788.93 & Ni~{\sc i} & 1.95 & 1.023e-02 & E93 & 102 & ---\\
4722.16 & Zn~{\sc i} & 4.01 & 4.074e-01 & BG80 & --- & 40 \\
4810.53 & Zn~{\sc i} & 4.06 & 6.761e-01 & BG80 & 66 & 35 \\
4883.68 & Y~{\sc ii} & 1.08 & 1.175e+00 & SN96 & --  & 79  \\
5087.43 & Y~{\sc ii} & 1.08 & 6.761e-01 & SN96 & 92  & 57  \\
5200.41 & Y~{\sc ii} & 0.99 & 2.692e-01 & SN96 & 86  & 48  \\
5205.72 & Y~{\sc ii} & 1.03 & 4.571e-01 & SN96 & --- & 62  \\
5289.81 & Y~{\sc ii} & 1.03 & 1.412e-02 & VWR  & 33  & --- \\
5402.78 & Y~{\sc ii} & 1.84 & 3.631e-01 & R03  & 49  & 15  \\
6127.46 & Zr~{\sc i} & 0.15 & 8.710e-02 & H82  & 15  & --- \\
6134.57 & Zr~{\sc i} & 0.00 & 5.248e-02 & B81  & 15  & --- \\
6143.18 & Zr~{\sc i} & 0.07 & 7.943e-02 & B81  & 15  & --- \\
4208.98 & Zr~{\sc ii}& 0.71 & 3.467e-01 & SN96 & --- &  68 \\
4317.32 & Zr~{\sc ii}& 0.71 & 4.169e-02 & SN96 & 33  &  23 \\ 
\hline
4816.50 & Zr~{\sc ii}& 1.01 & 1.000e-02 & VWR  & 28  &  10 \\  
5112.27 & Zr~{\sc ii}& 1.66 & 1.738e-01 & E93  & 49  &  19 \\
6496.90 & Ba~{\sc ii} & 0.60 & 4.170e-01 & WM80 & 220 & 176 \\
4934.83 & La~{\sc ii} & 1.25 & 1.191e-01 & VWR & 18  & --- \\ 
5303.53 & La~{\sc ii} & 0.32 & 4.467e-02 & VWR & 50  & 23  \\
5880.63 & La~{\sc ii} & 0.24 & 1.479e-02 & R04 & 14  & ---\\
6320.42 & La~{\sc ii} & 0.17 & 3.030e-02 & VWR & 60 & 22 \\
6390.48 & La~{\sc ii} & 0.32 & 2.750e-02 & S96  & 54 & 25 \\
6774.33 & La~{\sc ii} & 0.12 & 1.778e-02 & S96  & 52 & 20 \\
5117.17 & Ce~{\sc ii} & 1.40 & 1.023e+00 & VWR  & 41 & --- \\
5187.45 & Ce~{\sc ii} & 1.21 & 1.995e+00 & VWR  & 55 & 30 \\
5274.24 & Ce~{\sc ii} & 1.28 & 2.450e+00 & VWR & 56 & 35 \\
5330.58 & Ce~{\sc ii} & 0.87 & 7.383e-01 & VWR & 44 & 24 \\
5472.30 & Ce~{\sc ii} & 1.25 & 1.650e-01 & VWR & 36 & 18 \\
6051.80 & Ce~{\sc ii} & 0.23 & 2.512e-02 & S96  & 25 & 10 \\
4811.34 & Nd~{\sc ii} & 0.06 & 9.660e-02 & VWR &  81 & 47 \\
4820.34 & Nd~{\sc ii} & 0.20 & 6.902e-02 & VWR & --- & 37 \\ 
4959.12 & Nd~{\sc ii} & 0.06 & 1.213e-01 & VWR &  84 & 53 \\ 
4989.95 & Nd~{\sc ii} & 0.63 & 2.377e-01 & VWR &  85 & 40 \\
5063.72 & Nd~{\sc ii} & 0.98 & 1.746e-01 & VWR &  37 & 17 \\
5130.59 & Nd~{\sc ii} & 1.30 & 1.259e+00 & SN96&  79 & 79 \\
5212.36 & Nd~{\sc ii} & 0.20 & 1.995e-01 & E93 &  71 & --- \\
5249.58 & Nd~{\sc ii} & 0.98 & 1.202e+00 & SN96&  74 & 45 \\  
5311.46 & Nd~{\sc ii} & 0.99 & 2.754e-01 & SN96&  47 & 16 \\
5319.81 & Nd~{\sc ii} & 0.55 & 4.467e-01 & SN96&  86 & 52 \\
5361.47 & Nd~{\sc ii} & 0.68 & 3.981e-01 & SN96&  78 & 33 \\
5416.38 & Nd~{\sc ii} & 0.86 & 1.047e-01 & VWR &  31 & --- \\
5431.54 & Nd~{\sc ii} & 1.12 & 3.491e-01 & VWR &  39 & --- \\
5442.26 & Nd~{\sc ii} & 0.68 & 1.259e-01 & SN96&  45 &  16 \\
5740.88 & Nd~{\sc ii} & 1.16 & 2.754e-01 & VWR &  32 & --- \\
5842.39 & Nd~{\sc ii} & 1.28 & 2.506e-01 & VWR &  24 & --- \\
\hline
\end{tabular}
\medskip
References to Table~\ref{tab3}:
BG80: Bi\'emont \& Godefroid (\cite{Biem80}); 
B81: Bi\'emont et~al. (\cite{Biem81});
D2002: Depagne et~al. (\cite{Depa02});
DS91: Drake \& Smith (\cite{Drak91});
E93: Edvardsson et~al. (\cite{Edva93});
GS: Gratton \& Sneden (\cite{Grat88});
H82: Hannaford et~al. (\cite{Hann82});
MFW: Martin et~al. (\cite{Mart88});
PS:  Preston \& Sneden (\cite{Pres01});
R03: Reddy et~al. (\cite{Redd03});
R04: Reyniers et~al. (\cite{Reyn04}); 
S86: Smith et~al. (\cite{Smit86});
S96: Smith et~al. (\cite{Smit96}); 
SN96: Sneden et~al. (\cite{Sned96});
VWR: \mbox{van Winckel} \& Reyniers (\cite{vanw00});
WSM: Wiese et~al. (\cite{Wies69}); 
WM80: Wiese \& Martin (\cite{Wies80}).
\end{table}